# Predicting the Risk of Rheumatoid Arthritis and Its Age of Onset through Modelling Genetic Risk Variants with Smoking

lan C. Scott<sup>1,2</sup>\*, Seth D. Seegobin<sup>2</sup>, Sophia Steer<sup>3</sup>, Rachael Tan<sup>1</sup>, Paola Forabosco<sup>4</sup>, Anne Hinks<sup>5</sup>, Stephen Eyre<sup>5</sup>, Ann W. Morgan<sup>6</sup>, Anthony G. Wilson<sup>7</sup>, Lynne J. Hocking<sup>8</sup>, Paul Wordsworth<sup>9</sup>, Anne Barton<sup>5</sup>, Jane Worthington<sup>5</sup>, Andrew P. Cope<sup>1</sup>, Cathryn M. Lewis<sup>2,10</sup>

1 Academic Department of Rheumatology, Centre for Molecular and Cellular Biology of Inflammation, King's College London, London, United Kingdom, 2 Department of Medical and Molecular Genetics, King's College London, London, United Kingdom, 3 Department of Rheumatology, King's College Hospital, London, United Kingdom, 4 Istituto di Genetica delle Popolazioni, Consiglio Nazionale delle Ricerche, Sassari, Italy, 5 Arthritis Research UK Epidemiology Unit, Centre for Musculoskeletal Research, Institute of Inflammation and Repair, The University of Manchester, Manchester, United Kingdom, 6 Division of Musculoskeletal Disease, Leeds Institute of Molecular Medicine, University of Leeds and National Institute for Health Research – Leeds Musculoskeletal Biomedical Research Unit, Leeds, United Kingdom, 7 Academic Unit of Rheumatology, Department of Infection and Immunity, University of Sheffield Medical School, Sheffield, United Kingdom, 9 NIHR Oxford Musculoskeletal BRU, Nuffield Orthopaedic Centre, Oxford, United Kingdom, 10 Social, Genetic and Developmental Psychiatry Centre (MRC), Institute of Psychiatry, London, United Kingdom

# Abstract

The improved characterisation of risk factors for rheumatoid arthritis (RA) suggests they could be combined to identify individuals at increased disease risks in whom preventive strategies may be evaluated. We aimed to develop an RA prediction model capable of generating clinically relevant predictive data and to determine if it better predicted younger onset RA (YORA). Our novel modelling approach combined odds ratios for 15 four-digit/10 two-digit HLA-DRB1 alleles, 31 single nucleotide polymorphisms (SNPs) and ever-smoking status in males to determine risk using computer simulation and confidence interval based risk categorisation. Only males were evaluated in our models incorporating smoking as eversmoking is a significant risk factor for RA in men but not women. We developed multiple models to evaluate each risk factor's impact on prediction. Each model's ability to discriminate anti-citrullinated protein antibody (ACPA)-positive RA from controls was evaluated in two cohorts: Wellcome Trust Case Control Consortium (WTCCC: 1,516 cases; 1,647 controls); UK RA Genetics Group Consortium (UKRAGG: 2,623 cases; 1,500 controls). HLA and smoking provided strongest prediction with good discrimination evidenced by an HLA-smoking model area under the curve (AUC) value of 0.813 in both WTCCC and UKRAGG. SNPs provided minimal prediction (AUC 0.660 WTCCC/0.617 UKRAGG). Whilst high individual risks were identified, with some cases having estimated lifetime risks of 86%, only a minority overall had substantially increased odds for RA. High risks from the HLA model were associated with YORA (P<0.0001); ever-smoking associated with older onset disease. This latter finding suggests smoking's impact on RA risk manifests later in life. Our modelling demonstrates that combining risk factors provides clinically informative RA prediction; additionally HLA and smoking status can be used to predict the risk of younger and older onset RA, respectively.

Citation: Scott IC, Seegobin SD, Steer S, Tan R, Forabosco P, et al. (2013) Predicting the Risk of Rheumatoid Arthritis and Its Age of Onset through Modelling Genetic Risk Variants with Smoking. PLoS Genet 9(9): e1003808. doi:10.1371/journal.pgen.1003808

Editor: Greg Gibson, Georgia Institute of Technology, United States of America

Received March 27, 2013; Accepted August 5, 2013; Published September 19, 2013

**Copyright:** © 2013 Scott et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Funding:** This manuscript was undertaken as part of an Arthritis Research UK funded Clinical Research Fellowship (ICS; Grant Reference Number 19739). Funding for the WTCCC project was provided by the Wellcome Trust (http://www.wellcome.ac.uk) under award 076113, 085475 and 090355. This work made use of data and samples generated by the 1958 Birth Cohort (NCDS) (BRIF4130). Access to these resources was enabled via the 58READIE Project funded by Wellcome Trust and G1001799). A full list of the financial, institutional and personal contributions to the development of the 1958 Birth Cohort Biomedical resource is available at http://www2.le.ac.uk/projects/birthcohort. ICS and APC are supported by Arthritis Research UK (http://www.arthritisresearchuk.org) funding as are AH, SE, JW and AB (Grant Reference Number 17552). We also acknowledge support from the National Institutes of Health Research (http://www.nihr.ac.uk/Pages/default.aspx) Biomedical Research Centre at Guy's and St Thomas' NHS Foundation Trust in partnership with King's College London, from whom SDS receives funding. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

\* E-mail: ian.scott@kcl.ac.uk

## Introduction

Rheumatoid arthritis (RA) is a common chronic inflammatory disorder. It results in substantial morbidity and disability alongside high medical and societal costs [1], [2]. There is therefore growing interest in preventing its development. Such prevention requires an ability to reliably predict who will develop RA. Advances in characterising genetic and environmental risk factors for RA together with developments in modelling methodology make predicting its development a realistic possibility.

RA is a clinical syndrome spanning multiple subsets [3]. The commonest subdivision is by the presence or absence of rheumatoid factor (RF)/anti-citrullinated protein antibodies (ACPA), termed seropositive and seronegative RA respectively.

## **Author Summary**

Rheumatoid arthritis (RA) is a common, incurable disease with major individual and health service costs. Preventing its development is therefore an important goal. Being able to predict who will develop RA would allow researchers to look at ways to prevent it. Many factors have been found that increase someone's risk of RA. These are divided into genetic and environmental (such as smoking) factors. The risk of RA associated with each factor has previously been reported. Here, we demonstrate a method that combines these risk factors in a process called "prediction modelling" to estimate someone's lifetime risk of RA. We show that firstly, our prediction models can identify people with very high-risks of RA and secondly, they can be used to identify people at risk of developing RA at a younger age. Although these findings are an important first step towards preventing RA, as only a minority of people tested had substantially increased disease risks our models could not be used to screen the general population. Instead they need testing in people already at risk of RA such as relatives of affected patients. In this context they could identify enough numbers of high-risk people to allow preventive methods to be evaluated.

Risk factor evaluation has mainly focussed on seropositive RA with nearly half its genetic architecture known. *HLA-DRB1* alleles, in particular those encoding the shared epitope, dominate genetic risk accounting for approximately 36% of heritability [4]; 45 non-HLA variants explain approximately 15% of heritability [4]. Smoking is the main environmental risk factor [5]; it predisposes to seropositive RA and has a synergistic relationship with the shared epitope [6], [7]. Although single factors do not provide sufficient risk stratification, combining multiple factors within a prediction model may identify clinically relevant high- and lowrisk groups. The large risks conferred by HLA make such modelling an attractive prospect in RA despite limited success in other complex disorders [8–10].

RA develops over many years prior to clinical presentation [11]. Initially, individuals with genetic susceptibility variants are exposed to environmental risks; some may develop autoantibodies (RF/ACPA) [12]. A proportion will subsequently develop arthralgia, which may progress to an unclassified arthritis followed by a fully expressed RA phenotype. Pilot studies in unclassified arthritis indicate that secondary prevention may be possible with corticosteroids [13], [14], methotrexate [15] and biologics [16] attenuating the progression to RA. Although preventive treatments may be more effective before immune dysregulation and symptoms develop, primary prevention is not currently possible as no reliable method exists to identify asymptomatic high-risk individuals.

Prevention is likely to have a larger impact in younger onset RA (YORA) due to the increased health costs associated with a longer disease duration [17]. Genetic susceptibility factors may influence RA's age of onset with *HLA-DRB1*\*04 alleles [18–21] and multiple single nucleotide polymorphisms (SNPs) such as those tagging *VEGFA* [22], *RANKL* [19], [23], *MMP1-3* [22] and *PTPN22* [24], [25] loci associating with YORA.

One group has published two reports outlining predictive models for RA. Their models, built using 8 HLA alleles, 14–31 SNPs and clinical factors, generated an aggregate weighted genetic risk score (wGRS) formed from the product of individual-locus odds ratios (ORs) [26], [27]. They were reasonably accurate at determining disease status in approximately 1,200 cases and 1,200

controls, with a maximal area under the curve (AUC) of 0.752. They also demonstrated a better ability to predict erosive RA (a more severe phenotype). However, only a minority of the studied populations had significantly elevated risks for RA.

We report an alternative modelling approach to predicting RA. Our novel modelling method uses computer simulation to categorise risk profiles; our models also incorporate a larger number of HLA risk variants. The risk factors included in our modelling comprise 15 four-digit/10 two-digit HLA-DRB1 alleles, 31 SNPs and male ever-smoking status (as ever-smoking is a significant risk for RA in males only). We applied our models to two large cohorts of European ancestry: the Wellcome Trust Case Control Consortium (WTCCC) and the UK RA Genetics Group (UKRAGG) Consortium. Our primary aim was to determine if our approach would generate clinically relevant predictive values. Our secondary aim was to determine if our modelling better identified YORA. We demonstrate that clinically informative RA risk prediction is possible and that the risk of younger and older onset RA can be predicted using information on HLA and smoking status, respectively.

## **Materials and Methods**

#### **Ethics Statement**

All participants in WTCCC and UKRAGG were recruited after providing informed consent. UKRAGG was approved by the North West Multi-Centre Research Ethics Committee (MREC 99/8/84). Authors gained written permission and approval from WTCCC to undertake this work in the publically available WTCCC1 collections.

# **Study Populations**

The WTCCC dataset contains SNP data on 1,999 RA cases and 3,004 controls [28]. Controls were obtained from the 1958 British Birth Cohort and UK Blood Services. Genotyping was performed on the Affymetrix GeneChip 500k Mapping Array Set. Quality control (QC) procedures were undertaken excluding individuals with <97% SNP call rates, high heterozygosity, non-European ancestry or relatedness, discordance between genotype and phenotype data and duplicate samples. In the post-QC dataset information was available on 490,031 SNP markers; the total genotyping rate was 1.00. Two- or four-digit resolution *HLA-DRB1* tissue typing data were available on 1,837 cases and 1,647 controls.

The UKRAGG dataset contains SNP data on 5,024 RA cases and 4,281 controls from 6 UK centres [29]. Genotyping was performed using the Sequenom platform. Four hundred and four SNPs were genotyped over 8 staggered plexes; for each plex separate QC was undertaken excluding individuals and SNPs with <90% data present. In the post-QC dataset total genotyping rates were 0.73 owing to systematic differences in samples run on each plex. Two- or four-digit resolution *HLA-DRB1* tissue typing data were available on 3,420 cases and 1,500 controls.

Both datasets contained cases fulfilling the 1987 ACR classification criteria for RA [30]. *HLA-DRB1* tissue typing was undertaken (at two-digit or four-digit resolution) at individual centres, using commercially available semiautomated polymerase chain reaction-sequence-specific oligonucleotide probe (PCR-SSOP) typing techniques (or research assays based on PCR-SSOP linear array technology) [29]. Two-digit typing includes the allele group (Field 1) only; four-digit typing includes both the allele group and the allele subtype encoding a specific HLA protein (Field 2) (http://hla.alleles.org/nomenclature/naming.html). Table 1. Clinical characteristics of WTCCC/UKRAGG cases and controls included in modelling.

		WTCCC		UKRAGG		
		RA (n = 1,516)	Controls (n = 1,647)	RA (n = 2,623)	Controls (n = 1,500)	
Gender	Female	1,151 (76.0)	739 (50.0)	1,868 (71.2)	890 (59.9)	
RA Characteristics	RF+	1,452 (96.1)	-	2,385 (93.1)	-	
	ACPA+	1,061 (86.5)	-	1,508 (84.8)	-	
	Mean Age Of Onset (95% CI)	45.3 (44.6–46.1)	-	48.0 (47.5–48.6)	-	
	Erosive Disease	1,009 (71.1)	-	830 (69.7)	-	
	Nodules	-	-	859 (38.3)	-	
Smoking Status	Male Ever-Smokers	231 (80.5) <sup>a</sup>	422 (57.1) <sup>a</sup>	417 (78.8) <sup>a</sup>	149 (46.3) <sup>a</sup>	
	Female Ever-Smokers	552 (58.3) <sup>b</sup>	425 (57.7) <sup>b</sup>	758 (55.9) <sup>b</sup>	238 (39.3) <sup>b</sup>	

Data are number (%) unless otherwise stated. The following data are missing from WTCCC: gender in 2 cases and 169 controls; RF status in 5 cases; ACPA status in 290 cases; age of onset missing/inaccurate in 63 cases; erosive status in 96 cases; smoking status in 76 male cases, 204 female cases and 3 female controls. The following data are missing from UKRAGG: gender in 14 controls; RF status in 60 cases; ACPA status in 844 cases; age of onset missing/inaccurate in 93 cases; erosive status in 1,432 cases; nodular status in 378 cases; smoking status in 226 male cases, 513 female cases, 274 male controls and 284 female controls.

<sup>a</sup> = % of males that are ever smokers;

<sup>b</sup>=% of females that are ever smokers.

doi:10.1371/journal.pgen.1003808.t001

We undertook prediction modelling in seropositive cases and controls with *HLA-DRB1* tissue typing data available with or without additional SNP and smoking data (as most replicated risk loci are for seropositive RA and genetic risk is dominated by HLA) [4], [31]. The final cohorts comprised 1,516 cases and 1,647 controls from WTCCC and 2,623 cases and 1,500 controls from UKRAGG (Table 1).

#### Prediction Modelling Overview

Our modelling was performed within the R package, REGENT (Risk Estimation for Genetic and Environmental Traits), developed within our unit. This program incorporates published geneenvironment risk factor and disease statistics to categorise risk using a confidence interval (CI)-based approach within a simulated population. The methodology underlying REGENT has previously been described in detail [32], [33].

Genetic and environmental risk factors for input into REGENT are selected from the literature. Genetic risk factors require allelic ORs, allele frequencies, and sample sizes from relevant studies, in order to estimate precision. Environmental risk factors require ORs, standard errors and the proportion of the population exposed to the risk factor. Data on these risk factors are entered into REGENT as summary statistic input files, which are processed in two stages: the first develops the prediction model and the second runs the prediction model in real life data.

In the first stage REGENT simulates a population-distribution of disease risk. Risk profiles are simulated based on the frequency of each risk factor in the general population. Summary ORs for each risk profile are generated through combining the ORs for each genetic and environmental risk factor in a multiplicative model that assumes risk factor independence. CIs are generated using information on the variability of genetic risk factors (derived from the sample size of the risk variant discovery cohort) and environmental risk factors (standard error of the effect size). Each simulated risk profile's OR is initially calculated relative to a profile with no risk factors present; these are subsequently adjusted to ensure correct disease prevalence in the population, assigning a risk profile with a mean OR as having a baseline risk of 1.0. CIs are used to classify risk profiles into four risk categories (reduced, average, elevated and high-risk). Starting with the risk profile of baseline risk (OR = 1.0), any risk profile whose CI overlaps with this baseline CI is classified as being of average-risk (as this profile is not statistically different from baseline). Any risk profile whose CI resides fully below the baseline CI is classified as reduced-risk. Profiles with CIs above the baseline CI are classified as elevatedrisk. Furthermore, a high-risk group is determined by profiles whose CIs reside completely above the CI of the first risk profile classified as elevated-risk. An example of how this process is undertaken in a simplified model using 3 SNPs is provided in Figure S1.

In the second stage REGENT applies this simulated population profile to individual level data. Genotypes and environmental risk factor exposure data on each individual in the dataset of interest (WTCCC and UKRAGG) are entered into REGENT, which generates two measures of disease risk. Firstly, each individual's summary OR (95% CI) for RA is calculated (relative to the baseline individual with an OR of 1.0); as with the simulated population, risk factors are combined in a multiplicative model. This summary OR informs the individual of their risk of developing RA. Secondly, each individual is assigned a risk category for RA. This is undertaken through comparing the CI of each individual's summary OR to those of the simulated risk distribution in the same manner as described in stage 1. This risk category informs an individual whether they are at an increased or reduced risk of disease, relative to the average person in the general population.

## Prediction Model Components Identified from Meta-Analyses

**Genetic Risk Factors.** We identified genetic susceptibility variants for potential inclusion in our prediction modelling from two large, recently published meta-analyses [34], [35]. We sought to include only susceptibility alleles attaining genome-wide significance ( $P_{GWAS} < 5 \times 10^{-8}$ ); this ensured that the alleles modelled were replicated RA genetic risk factors. These comprised 15 four-digit and 10 two-digit *HLA-DRB1* alleles and 35 non-HLA SNPs.

**Environmental Risk Factor.** We included the environmental risk factor smoking in our modelling. Other factors proposed to influence RA risk such as alcohol were not included: firstly the evidence underlying these is uncertain, with associations often present in case-control and not cohort studies [36] and secondly We used published ORs from the most recent meta-analysis evaluating smoking as an RA risk factor [5]. In this meta-analysis ever-smoking was a significant risk for seropositive RA in males only (OR 3.02; 95% CI 2.35–3.88) with a substantially smaller and non-significant (CIs contain 1) impact seen in females (OR 1.34; 95% CI 0.99–1.80). We therefore hypothesized that smoking would not improve prediction in women (confirmed in preliminary analyses; Table S1). As a result only males were evaluated in our modelling incorporating ever-smoking.

Although smoking interacts with the shared epitope we did not factor this into our modelling. This is because studies reporting summary ORs for this interaction [6], [29], [37], [38] have marked heterogeneity between them; therefore using meta-analysis techniques to obtain pooled ORs for shared epitope-smoking status combinations would be inaccurate and thus inappropriate. Examples of this heterogeneity include: (1) studies reporting risks stratified by different smoking levels, which would require an inverse variance fixed-effects model to obtain common ORs for all smokers within studies in addition to a random-effects model to estimate pooled ORs across studies; (2) two studies classifying the shared epitope at two-digit resolution, thus incorporating nonshared epitope alleles [6], [37]; (3) two studies not including all known shared epitope alleles [29], [38].

# Prediction Model Component Availability in WTCCC and UKRAGG

Two-digit or four-digit *HLA-DRB1* tissue typing data were available in all evaluated individuals. In WTCCC 1,342 seropositive cases, 966 ACPA-positive cases and 1,126 controls had fourdigit resolution data available on both alleles; 29 seropositive cases, 14 ACPA-positive cases and 159 controls had two-digit resolution data available on both alleles; 145 seropositive cases, 81 ACPApositive cases and 362 controls had mixed-digit resolution data (one *HLA-DRB1* allele known at four-digit and the other at twodigit resolution) available. In UKRAGG 1,534 seropositive cases, 1,108 ACPA-positive cases and 735 controls had four-digit resolution data available on both alleles; 312 seropositive cases, 66 ACPA-positive cases and 205 controls had two-digit resolution data available on both alleles; 777 seropositive cases, 334 ACPApositive cases and 560 controls had mixed-digit resolution data available.

We excluded 4 SNPs attaining  $P_{GWAS}$  in the meta-analysis for the following reasons: 1 (rs11676922) was in high linkage disequilibrium (r<sup>2</sup>>0.9; HapMap release 22 CEU population panel) [39] with another (rs10865035) – in this case the latter SNP was included due to a previous association with RA – and 3 SNPs/ proxy SNPs were unavailable (rs10488631, rs6859219 and rs934734 in UKRAGG; rs6822844, rs874040 and rs951005 in WTCCC). Eleven and two proxy SNPs were used in WTCCC and UKRAGG respectively (Table S2) [39].

Data on ever-smoking status were available in 287 male cases and 739 male controls in WTCCC and 529 male cases and 322 male controls in UKRAGG.

## **Final Prediction Models**

To examine the contribution of each gene-environment component to prediction we constructed several models. These comprised a SNP model (with 31 SNPs), an HLA model (10 twodigit and 15 four-digit *HLA-DRB1* alleles), an HLA-SNP model (combining HLA and SNP model components), an HLA-smoking model (combining *HLA-DRB1* alleles with ever-smoking status) and an HLA-SNP-smoking model (combining *HLA-DRB1* alleles, 28 SNPs and ever-smoking status). Only the 28 SNPs present in both WTCCC and UKRAGG were incorporated in the last model. The latter two models, which included smoking, were evaluated in males only.

The decision to combine two-digit and four-digit *HLA-DRB1* alleles in the HLA model was undertaken to avoid removing the substantial number of individuals with mixed resolution typing. Preliminary analyses confirmed the validity of this approach with no significant differences seen in the discriminative abilities of HLA models incorporating (1) two-digit alleles only; (2) four-digit alleles only and (3) a mixed resolution of alleles (Table S3). Within our mixed resolution modelling the risks for each HLA allele were included only once per individual at the highest resolution at which they were known.

Only individuals with available data on relevant risk factors were included in models incorporating those risk factors. Therefore only males with available smoking data were included in the HLA-smoking and HLA-SNP-smoking models. Similarly only individuals with data available on the modelled SNPs could be included in the HLA-SNP and HLA-SNP-smoking models. Owing to missing data the number of individuals evaluated in each prediction model fell as more risk factors were included (Figure 1).

#### Statistical Analyses

**Evaluating Dataset Validity.** To compare the representativeness of our datasets to published RA populations we summarised clinical features of cases and controls (Table 1) and calculated effect allele frequencies and allelic ORs (95% CIs) (Tables 2 and 3). For the *HLA-DRB1* allele case-control association analysis (Table 2) the two-digit resolution allele results included both individuals with two-digit resolution typing and collapsed four-digit resolution typing. This approach was undertaken due to the small number of individuals with two-digit typing data in WTCCC/UKRAGG. The meta-analysis from which we obtained our risk alleles had almost identical allele frequencies when comparing two-digit alleles and four-digit alleles collapsed to twodigit resolution [35]; comparing our datasets to the meta-analysis findings in this manner was therefore appropriate.

**Comparing Model Classification Abilities.** To evaluate the ability of each model to correctly classify disease status we constructed receiver operating characteristic (ROC) curves and measured the AUC; this is established methodology in determining genetic classification test efficacy [40], [41]. Higher AUCs indicate better classification. An AUC>0.5 signifies some discriminative ability; a perfect classifier has an AUC of 1. AUCs were calculated and compared using DeLong's method [42] performed within the R package, pROC [43].

**Comparing Model Generated Risk Distributions.** The risk distributions for cases and controls under each model were compared by plotting the logarithmic OR for seropositive RA for each individual ordered by risk.

**Calculating Lifetime Risk of RA.** Due to the low prevalence of RA [44], ORs approximate relative risks [45]. Therefore to calculate lifetime risks of seropositive RA we multiplied published lifetime risks by the summary OR for RA generated by our prediction models. As UK lifetime risks of RA are unknown we used estimates from a large US cohort study (2.4% for women; 1.1% for men) [46].

**Evaluating YORA Prediction.** The role of HLA, SNPs and ever-smoking status in determining age of RA onset was evaluated using individual-level OR outputs from the REGENT models in a Cox univariate analysis with gender, smoking status and smoking status-gender interaction used as covariates. Factors indicated as likely predictors of age of onset were then examined



Figure 1. Number of individuals evaluated in each prediction model. doi:10.1371/journal.pgen.1003808.g001

simultaneously in a multivariate analysis incorporating backward elimination of non-significant factors (P > 0.05). We found no evidence of a "gender-smoking interaction" effect on the age of RA onset in either dataset (WTCCC P=0.0823 and UKRAGG P=0.8369; Table 4). This excluded a significant influence of gender on the relationship between smoking and the age at which RA developed. We therefore included both sexes when evaluating smoking's effect on the age of onset. Proportional hazards assumptions were verified using visual inspection of log-log plots [47]. To further demonstrate associations between significant factors and age of onset we constructed Kaplan-Meier estimates of the cumulative risk for cases, stratified by REGENT risk categorisation from the relevant models, alongside the presence/ absence of other risk factors. We used a Cox multivariate approach to establish which four-digit HLA-DRB1 alleles influenced age of onset (fitting all alleles simultaneously using stepwise selection, removing non-significant alleles from the final model). All time to event analyses were performed using SAS version 9.3 (SAS Institute, Cary, NC).

#### Separate Analyses for ACPA-Positive RA

We undertook modelling separately for seropositive (RF and/or ACPA present) RA and ACPA-positive RA since *HLA-DRB1* allelic ORs were obtained from a meta-analysis evaluating ACPA-positive RA [35], and the shared epitope alleles, non-HLA SNPs and smoking predominantly associate with ACPApositive disease [4], [48–50]. We therefore hypothesised our modelling would perform better for ACPA-positive RA. As this was confirmed in the risk categorisation results we restricted further analyses (AUC and lifetime risk calculations, examining modelling associations with age of RA onset) to ACPA-positive RA.

# Results

# **Dataset Validity**

**Genetic Risk Factors.** In both WTCCC and UKRAGG the effect allele frequencies and ORs for seropositive RA were generally similar to published data (Tables 2 and 3). Exceptions occurred at the four-digit *HLA-DRB1* alleles \*04:08 and \*15:01 (absent from controls in WTCCC and UKRAGG respectively), at \*01:01, \*11:01, \*11:04, \*13:01 and \*15:01 in WTCCC and \*08:01 in UKRAGG (significantly lower allele frequencies in controls than expected). The absence of \*04:08 in controls was probably a chance finding since it has a frequency of 0.005. The remaining discrepancies resulted from lower four-digit tissue typing rates for these alleles in controls, which were more often typed at two-digits, compared with cases. Although this could introduce bias,

Table 2. Classical HLA-DRB1 allele frequencies and their association with seropositive RA in WTCCC and UKRAGG.

	Published Meta-Analysis [35]			wтссс			UKRAGG		
HLA-DRB1 Allele	OR (95% CI)	MAF Co	MAF Ca	OR (95% CI)	MAF Co	MAF Ca	OR (95% CI)	MAF Co	MAF Ca
*01	1.30 (1.21–1.40)	0.113	0.145	1.53 (1.31–1.78)	0.104	0.151	1.27 (1.11–1.45)	0.121	0.149
*01:01	1.38 (1.28–1.50)	0.097	0.133	5.88 (4.62–7.55)	0.026	0.136	1.25 (1.06–1.47)	0.081	0.099
*03	0.59 (0.54–0.64)	0.128	0.082	0.67 (0.58–0.78)	0.148	0.105	0.76 (0.67–0.86)	0.159	0.125
*03:01	0.59 (0.54–0.64)	0.128	0.082	0.65 (0.55–0.76)	0.145	0.099	0.44 (0.37–0.51)	0.130	0.061
*04	3.71 (3.49–3.93)	0.174	0.450	2.90 (2.59–3.24)	0.213	0.439	3.19 (2.86–3.56)	0.184	0.419
*04:01	4.14 (3.86–4.44)	0.104	0.309	2.93 (2.57–3.35)	0.124	0.293	3.00 (2.63-3.42)	0.111	0.272
*04:04	3.17 (2.83–3.54)	0.036	0.091	1.86 (1.52–2.28)	0.052	0.092	2.56 (2.08-3.18)	0.039	0.093
*04:05	2.31 (1.77–3.01)	0.007	0.012	2.01 (1.12–3.73)	0.006	0.012	2.61 (1.34–5.58)	0.004	0.010
*04:08	5.48 (4.11-7.30)	0.005	0.017	_a	0.000	0.021	2.78 (1.70-4.76)	0.007	0.018
*07	0.49 (0.45–0.54)	0.133	0.064	0.48 (0.41-0.56)	0.154	0.080	0.54 (0.46–0.62)	0.142	0.081
*07:01	0.49 (0.45–0.54)	0.133	0.064	0.41 (0.35–0.49)	0.154	0.070	0.37 (0.32-0.44)	0.140	0.057
*08	0.41 (0.34–0.50)	0.029	0.013	0.39 (0.24–0.62)	0.022	0.009	0.30 (0.20-0.44)	0.029	0.009
*08:01	0.34 (0.26-0.44)	0.019	0.009	0.27 (0.13–0.53)	0.014	0.004	0.69 (0.33–1.46)	0.005	0.003
*10	2.53 (2.04–3.14)	0.008	0.020	1.97 (1.11–3.59)	0.006	0.012	1.75 (1.04–3.07)	0.007	0.012
*10:01	2.53 (2.04–3.14)	0.008	0.020	1.97 (1.11–3.59)	0.006	0.012	1.48 (0.85–2.67)	0.006	0.009
*11	0.48 (0.43-0.54)	0.094	0.039	0.50 (0.39–0.64)	0.064	0.033	0.42 (0.34–0.53)	0.065	0.028
*11:01	0.44 (0.38–0.52)	0.061	0.028	0.80 (0.55–1.14)	0.023	0.018	0.33 (0.23-0.47)	0.030	0.010
*11:04	0.15 (0.10-0.23)	0.024	0.008	0.79 (0.41–1.49)	0.008	0.006	0.38 (0.15–0.91)	0.005	0.002
*13	0.33 (0.30-0.37)	0.114	0.044	0.41 (0.33–0.50)	0.098	0.042	0.46 (0.38–0.55)	0.084	0.040
*13:01	0.28 (0.24–0.33)	0.061	0.021	0.77 (0.54–1.08)	0.026	0.020	0.42 (0.29–0.59)	0.027	0.011
*13:02	0.29 (0.23–0.38)	0.027	0.012	0.59 (0.38–0.90)	0.020	0.012	0.27 (0.18-0.42)	0.023	0.006
*14	0.50 (0.40–0.62)	0.025	0.012	0.51 (0.34–0.76)	0.024	0.013	0.45 (0.31–0.65)	0.023	0.010
*14:01	0.46 (0.36–0.59)	0.022	0.011	0.43 (0.28-0.66)	0.024	0.011	0.67 (0.34–1.33)	0.006	0.004
*15	0.59 (0.54–0.64)	0.142	0.092	0.63 (0.53–0.75)	0.128	0.084	0.60 (0.53–0.70)	0.146	0.093
*15:01	0.57 (0.53-0.62)	0.136	0.089	1.09 (0.87–1.37)	0.051	0.055	_ <sup>a</sup>	0.000	0.025

All alleles attained genome-wide significance in the published meta-analysis; MAF = minor allele frequency; Co = controls; Ca = Cases;

<sup>a</sup>=OR incalculable due to no allele copies in the control group.

doi:10.1371/journal.pgen.1003808.t002

especially in the context of case-control association analyses, we do not consider it significantly affected our prediction modelling because these alleles were incorporated in our models at both twodigit and four-digit resolution (in most cases in the reference metaanalysis the two-digit alleles had similar allele frequencies and ORs compared with the four-digit alleles) and our risks were obtained from an external source [35].

SNP discrepancies occurred at rs3761847 in WTCCC and rs26232 and rs540386 in UKRAGG, which had ORs in the opposite direction to published results although the dataset and meta-analysis 95% CI's overlapped for two SNPs. Additionally the minor allele frequencies (MAFs) in controls were similar to those expected. These discrepancies probably represent normal variation as opposed to systematic genotyping differences.

Most *HLA-DRB1* alleles had significant associations with RA, with only 4 (16%) alleles in WTCCC and 3 (12%) alleles in UKRAGG having 95% CIs containing 1.0. A substantial proportion of SNPs -13 (42%) in WTCCC and 15 (48%) in UKRAGG – had 95% CIs containing 1.0 reflecting their modest effect sizes, which required large discovery cohort sizes for detection.

**Environmental Risk Factors.** The ORs for seropositive RA in ever-smokers were 3.10 (95% CI 2.22-4.37) in WTCCC and 4.32 (95% CI 3.16-5.92) in UKRAGG for males and 1.02 (95% CI 0.84-1.25) in WTCCC and 1.96 (95% CI 1.61-2.40) in

UKRAGG for females. The meta-analysis gender discrepancy surrounding the effect of ever-smoking on RA risk [5] was therefore mirrored in our datasets supporting the inclusion of only males in our smoking models.

#### **Risk Prediction**

**Risk Categorisation.** As hypothesized, our modelling more accurately categorised ACPA-positive RA as high-risk compared with seropositive RA (Figure 2 and Table S4). The HLA model provided most prediction in both datasets, classifying approximately one third of ACPA-positive RA as high-risk and two thirds of controls reduced-risk. Although the SNP model provided some prediction it classified most individuals as average-risk, reflecting the overlapping CIs generated by including many risk factors of a small effect size.

In WTCCC, the full genetic (HLA-SNP) model performed slightly better than HLA alone. Additional smoking data conferred subtle improvements in categorisation; this is particularly seen with the HLA-SNP-smoking model, which classified over half of ACPA-positive RA elevated/high-risk and 59% of controls reduced-risk.

In UKRAGG the addition of SNPs to HLA alleles increased the average-risk group size with no clear predictive benefits. The incorporation of smoking substantially improved prediction: the Table 3. Non-HLA RA susceptibility SNP allele frequencies and their association with seropositive RA in WTCCC and UKRAGG.

	Published Meta-Analysis [34]			WTCCC		UKRAGG		
Loci	SNP	MAF <sup>a</sup>	OR	MAF Ca/Co	OR (95% CI)	MAF Ca/Co	OR (95% CI)	
PTPN22	rs2476601	0.10	1.94 (1.81–2.08)	0.18/0.10	2.02 (1.73–2.36)	0.16/0.10	1.60 (1.38–1.85)	
TNFAIP3	rs6920220	0.22	1.22 (1.16–1.29)	0.27/0.23	1.26 (1.12–1.41)	0.25/0.21	1.29 (1.15–1.44)	
ANKRD55, IL6ST	rs6859219	0.21	0.78 (0.72–0.85)	0.17/0.20	0.80 (0.70–0.91)	-	-	
CD40	rs4810485	0.25	0.85 (0.80-0.90)	0.22/0.24	0.87 (0.77–0.99)	0.22/0.25	0.83 (0.74–0.93)	
CTLA4	rs3087243	0.44	0.87 (0.83-0.91)	0.43/0.44	0.95 (0.86–1.06)	0.43/0.47	0.86 (0.78–0.94)	
TNFAIP3	rs5029937	0.04	1.40 (1.24–1.58)	0.06/0.04	1.58 (1.24–2.02)	0.05/0.04	1.39 (1.06–1.82)	
IL2RA	rs706778	0.40	1.14 (1.09–1.20)	0.46/0.42	1.17 (1.05–1.29)	0.43/0.40	1.13 (1.02–1.25)	
RBPJ	rs874040	0.30	1.14 (1.08–1.20)	-	-	0.33/0.31	1.11 (1.00–1.23)	
TRAF1, C5	rs3761847	0.43	1.13 (1.08–1.18)	0.45/0.46	0.96 (0.87–1.07)	0.46/0.43	1.12 (1.01–1.24)	
STAT4	rs7574865	0.22	1.16 (1.10–1.23)	0.21/0.19	1.12 (0.99–1.27)	0.25/0.22	1.18 (1.05–1.32)	
SPRED2	rs934734	0.49	1.13 (1.08–1.19)	0.53/0.51	1.11 (1.00–1.23)	-	-	
CCR6	rs3093023	0.43	1.13 (1.08–1.19)	0.42/0.40	1.10 (0.99–1.22)	0.47/0.44	1.16 (1.05–1.28)	
РХК	rs13315591	0.09	1.29 (1.17–1.43)	0.10/0.09	1.11 (0.94–1.32)	0.08/0.07	1.10 (0.91–1.33)	
C5orf30	rs26232	0.32	0.88 (0.84–0.93)	0.34/0.40	0.78 (0.70–0.86)	0.31/0.31	1.03 (0.92–1.14)	
CCL21	rs951005	0.16	0.84 (0.78–0.90)	-	-	0.13/0.15	0.86 (0.75–0.99)	
REL	rs13031237	0.37	1.13 (1.07–1.18)	0.45/0.43	1.07 (0.96–1.18)	0.41/0.37	1.22 (1.10–1.35)	
AFF3	rs10865035	0.47	1.12 (1.07–1.17)	0.50/0.46	1.19 (1.07–1.31)	0.48/0.45	1.16 (1.05–1.27)	
PRKCQ	rs4750316	0.19	0.87 (0.82–0.92)	0.16/0.20	0.77 (0.67–0.87)	0.18/0.19	0.89 (0.79–1.00)	
RF5	rs10488631	0.11	1.19 (1.10–1.28)	0.12/0.10	1.22 (1.04–1.44)	-	-	
TNFRSF14	rs3890745	0.32	0.89 (0.85-0.94)	0.29/0.32	0.85 (0.76–0.95)	0.32/0.33	0.97 (0.88–1.08)	
CD2, CD58	rs11586238	0.24	1.13 (1.07–1.19)	0.26/0.24	1.08 (0.96–1.21)	0.26/0.26	1.05 (0.93–1.17)	
BLK	rs2736340	0.25	1.12 (1.07–1.18)	0.27/0.25	1.10 (0.98–1.24)	0.26/0.24	1.14 (1.01–1.28)	
CD28	rs1980422	0.24	1.12 (1.06–1.18)	0.25/0.23	1.13 (1.01–1.28)	0.26/0.23	1.15 (1.03–1.30)	
PRDM1	rs548234	0.33	1.10 (1.05–1.16)	0.36/0.34	1.11 (1.00–1.23)	0.35/0.35	1.00 (0.90–1.10)	
CCL21	rs2812378	0.34	1.10 (1.05–1.16)	0.38/0.34	1.17 (1.05–1.30)	0.36/0.35	1.02 (0.92–1.13)	
PTPRC	rs10919563	0.13	0.88 (0.82–0.94)	0.11/0.13	0.82 (0.70–0.95)	0.13/0.14	0.93 (0.80–1.07)	
KIF5A, PIP4K2C	rs1678542	0.38	0.91 (0.87–0.96)	0.34/0.37	0.86 (0.77–0.95)	0.35/0.35	0.97 (0.88–1.07)	
TRAF6	rs540386	0.14	0.88 (0.83-0.94)	0.11/0.13	0.90 (0.77–1.05)	0.14/0.13	1.03 (0.89–1.19)	
CGR2A	rs12746613	0.12	1.13 (1.06–1.21)	0.14/0.12	1.17 (1.01–1.36)	0.14/0.11	1.26 (1.08–1.46)	
TAGAP	rs394581	0.30	0.91 (0.87-0.96)	0.28/0.30	0.92 (0.82–1.03)	0.28/0.29	0.94 (0.84–1.05)	
TNFAIP3	rs10499194	0.27	0.91 (0.87–0.96)	0.25/0.27	0.90 (0.80–1.01)	0.26/0.28	0.90 (0.81-1.00)	
L2, IL21	rs6822844	0.18	0.90 (0.84-0.95)	-	-	0.15/0.19	0.80 (0.71-0.91)	
IL2RA	rs2104286	0.27	0.92 (0.87–0.97)	0.24/0.27	0.85 (0.76–0.96)	0.25/0.26	0.94 (0.84–1.04)	
IL2RB	rs3218253	0.26	1.09 (1.03–1.15)	0.29/0.25	1.22 (1.09–1.37)	0.29/0.27	1.09 (0.98–1.22)	

SNPs are ordered by significance (most significant by  $P_{GWAS}$  listed first); all alleles attained genome-wide significance in the published meta-analysis; Ca = Cases; Co = Controls; MAF = Minor Allele Frequency;

<sup>a</sup> = MAF in controls.

doi:10.1371/journal.pgen.1003808.t003

HLA-SNP-smoking model classified 38% ACPA-positive RA vs. 3% controls as high-risk and 70% controls vs. 18% ACPA-positive RA as reduced-risk.

The general trend of improved prediction through modelling increasing numbers of risk factors is highlighted by the ratios of the percentage of ACPA-positive cases to controls classified high-risk by each model. In WTCCC these comprise 3.4 for the SNP model, 3.8 for the HLA model, 5.8 for the HLA-SNP model, 4.8 for the HLA-smoking model and 6.0 for the HLA-SNP-smoking model. Similarly, the ratios of the percentage of controls to ACPA-positive cases classified reduced-risk in WTCCC comprise 2.3 for the SNP model, 2.4 for the HLA model, 3.4 for the HLA-SNP model, 4.0 for the HLA-smoking model and 4.7 for the HLA-SNP-smoking model. Similar findings were present in UKRAGG.

AUC Assessments. In WTCCC AUCs for the SNP, HLA, HLA-SNP, HLA-smoking and HLA-SNP-smoking models in discriminating between ACPA-positive RA and controls comprised 0.660 (95% CI 0.638–0.681), 0.764 (95% CI 0.746–0.782), 0.796 (95% CI 0.779–0.813), 0.813 (95% CI 0.784–0.841) and 0.837 (95% CI 0.810–0.865), respectively (Figure 3). Significant differences in AUCs were observed between all three genetic models: SNP and HLA models P<0.0001; HLA and HLA-SNP models P=0.0118. Smoking data significantly improved discrim-

Table 4. Relationship between modelling components and age of RA onset.

	WTCCC			UKRAGG					
		Univariate Analysis <sup>c</sup>			Univariate Analysis		Multivariate Analysis		
Modelling Component	No. Cases Examined	<i>P</i> -Value	Hazard Ratio (95% CI)	No. Cases Examined	<i>P</i> -Value	Hazard Ratio (95% CI)	<i>P</i> -Value	Hazard ratio (95% CI)	
HLA <sup>a,b</sup>	1022	< 0.0001	1.034 (1.018–1.050)	1456	0.0004	1.025 (1.011–1.038)	0.0003	1.026 (1.012–1.040)	
SNP <sup>a</sup>	1022	0.1804	1.043 (0.981–1.110)	284	0.294	1.075 (0.939–1.230)	-	-	
Gender <sup>b</sup>	1021	0.2157	0.914 (0.792–1.054)	1456	0.0107	0.864 (0.722–0.967)	0.0465	0.885 (0.786-0.998)	
Smoking <sup>b</sup>	962	0.1301	0.902 (0.789–1.031)	1361	0.0009	0.830 (0.743–0.927)	0.0041	0.848 (0.757–0.949)	
Gender-Smoking Interaction <sup>b</sup>	961	0.0823	0.870 (0.744–1.018)	1361	0.009	0.846 (0.746-0.959)	0.8369	-	

 $^{a}$  = HLA and SNP variables represent the summary OR scores generated by the models incorporating HLA and SNP data respectively;

<sup>b</sup> = variables included in UKRAGG multivariate model after variable pruning using backwards selection and model comparison with Akaike's Information Criterion; <sup>c</sup> = as only one parameter was significant in the WTCCC univariate analysis no multivariate model was fitted.

doi:10.1371/journal.pgen.1003808.t004

ination with differences observed between HLA and HLAsmoking model AUCs (P=0.0051) and HLA-SNP and HLA-SNP-smoking model AUCs (P=0.0120).

In UKRAGG AUCs for the SNP, HLA, HLA-SNP, HLAsmoking and HLA-SNP-smoking models in discriminating between ACPA-positive RA and controls comprised 0.617 (95% CI 0.577–0.656), 0.748 (95% CI 0.731–0.765), 0.756 (95% CI 0.723– 0.790), 0.813 (95% CI 0.782–0.845) and 0.857 (95% CI 0.804– 0.910), respectively (Figure 3). The HLA model had significantly better discrimination than the SNP model (P<0.0001). Combined SNP and HLA data did not improve discrimination with no differences observed between AUCs for the HLA and HLA-SNP models (P=0.665) or the HLA-smoking and HLA-SNP-smoking models (P=0.1671). Additional smoking information significantly improved modelling discrimination with significant differences observed between HLA and HLA-smoking model AUCs (P = 0.0003).

An overview of the main findings for each of the 5 prediction models, alongside the differences between them is provided in Figure S2.

**Risk Distributions.** In both datasets the HLA model provided most risk prediction generating substantially higher and lower ORs for RA in cases and controls respectively compared with the SNP model (Figure 4).

In WTCCC the addition of other risk factors to the *HLA-DRB1* alleles resulted in further small incremental increases in ORs for RA in cases; a less pronounced reduction in risk was seen in controls.



Figure 2. Risk categorisation of RA and controls by each prediction model. The y-axis on each graph refers to the proportion of cases/ controls in each risk category; cont = controls; sero+ = seropositive RA; ACPA+ = ACPA-positive RA. doi:10.1371/journal.pgen.1003808.g002



**Figure 3. Prediction model receiver operating characteristic curves.** Panel A = WTCCC; Panel B = UKRAGG; ROCs calculated for discriminating between ACPA-positive RA and controls; AUC = area under the curve. WTCCC model AUC comparisons: SNP versus HLA, P < 0.0001; HLA versus HLA-SNP, P = 0.0118; HLA-SNP versus HLA-Smoking, P = 0.3327; HLA-Smoking versus HLA-SNP-Smoking, P = 0.0001. UKRAGG model AUC comparisons: SNP versus HLA, P < 0.0001; HLA versus HLA-Smoking versus HLA-SNP-Smoking, P = 0.0145; HLA-SNP versus HLA-SNP-Smoking, P = 0.1671. doi:10.1371/journal.pgen.1003808.g003

In UKRAGG the addition of SNPs to HLA data provided no changes in case risk profiles, although a minority of controls had lower ORs. Additional smoking data resulted in significantly higher ORs for cases; only the HLA-SNP-smoking model clearly generated lower risk profiles for controls.

**Lifetime Risk Prediction.** Evaluating risks using genetics (HLA-SNP model) alone the highest risk WTCCC ACPA-positive case had an OR for seropositive RA of 79; as a male his lifetime risk was estimated at 86%. The highest risk control had an OR of 22; as a female her lifetime risk was estimated at 53%. Despite such high individual odds only a relative minority had relevant increased lifetime risks: using the same HLA-SNP model 49 (4.61%) ACPA-positive cases and 1 (0.07%) control had ORs for seropositive RA>20 (lifetime risks >48% if female and >22% if male) in WTCCC. In UKRAGG 9 (3.06%) ACPA-positive cases and 1 (0.17%) control had ORs>20.

The HLA-SNP-smoking model identified the greatest proportion of cases with substantially increased lifetime risks for RA. This

model identified 18 (7.53%) and 3 (3.75%) ACPA-positive male cases to have ORs for seropositive RA>20 (lifetime risk >22%) in WTCCC and UKRAGG respectively; no controls had ORs>20.

## Younger Onset RA Prediction

In WTCCC the HLA model summary OR score was the only significant predictor of age of RA onset (Table 4). The hazard ratio (HR) was 1.034 (P<0.0001), which indicated that the hazard (the rate at which RA occurred) was greater in individuals with higher HLA derived ORs than those with lower ORs. Therefore a higher HLA model generated risk score associated with RA occurring at a faster rate and thus YORA. Conversely ever-smoking was associated with older onset RA: the HR of 0.902 indicated a smaller hazard (RA occurred at a slower rate) in ever-smokers compared with never-smokers, although this was not significant (P=0.1301).

In UKRAGG the HLA model summary OR score, gender and smoking status were significant independent predictors of age of



**Figure 4. Prediction model generated risk profiles for ACPA-positive RA and controls.** Panel A = WTCCC; Panel B = UKRAGG; the upper set of lines for each model refer to RA cases; the lower set of lines refer to controls; OR = odds ratio. doi:10.1371/journal.pgen.1003808.q004

onset. An increasing HLA summary OR score associated with YORA (P=0.0003, HR 1.026); ever-smoking (P=0.0041, HR 0.848) and male gender (P=0.0465, HR 0.885) associated with older onset RA.

We considered that the non-significant relationship between smoking and age of onset in WTCCC reflected a limited sample size with our power to detect a 0.88 HR in the 962 WTCCC cases approximately 51% compared with 65% for the 1,361 UKRAGG cases. We therefore undertook a pooled analysis of both datasets (incorporating an additional "study" variable to account for dataset median age of onset differences). This confirmed that HLA derived risk scores significantly associated with YORA (P<0.0001, HR 1.030) and ever-smoking significantly associated with older onset RA (P=0.0489, HR 0.889).

Kaplan-Meier curves of age of onset stratified by HLA model risk categorisation further demonstrate the association of HLA risk profiles with YORA (Figure 5) with cases classified high-risk having significantly younger onset ages compared to those classified reduced-risk. In WTCCC the difference in the median time to RA (time point at which half the cases have developed RA) was 3 years between those classed high- and reduced-risk (Log-Rank = 11.43; P = 0.0007). In UKRAGG a stronger association was seen (Log-Rank = 27.33; P < 0.0001) with a difference in median time to RA onset between risk groups of 6 years. Further stratification by ever-smoking status demonstrated a trend towards an older onset age in ever-smokers. In WTCCC the median time to onset difference between high-risk never-smokers and reduced-risk ever-smokers was 7 years (Log-Rank = 14.42; P = 0.0024); a larger disparity was seen in UKRAGG with a difference of 12 years observed (Log-Rank = 46.2505; P < 0.0001).

Examining which four-digit resolution *HLA-DRB1* alleles influenced onset age revealed significant associations between age of onset and \*03:01 (P=0.0313), \*04:01 (P=0.0001), \*04:08 (P=0.0032) and \*13:02 (P=0.0097) in WTCCC and \*04:01 (P<0.0001) and \*04:04 (P=0.0243) in UKRAGG. Three of these alleles (\*04:01, \*04:04 and \*04:08) are shared-epitope alleles.

#### Discussion

We have demonstrated that predicting RA development is possible with our prediction models able to identify individuals with clinically relevant increased risks for seropositive RA. Our modelling indicates that most prediction is provided by *HLA*-*DRB1* alleles and, to a lesser extent, smoking in males; non-HLA susceptibility SNPs provide only minor predictive benefits. These findings are consistent with the estimations of heritability variance



Figure 5. Kaplan-Meier curves: RA age of onset stratified by HLA model risk categorisation and smoking status. Panel A=WTCCC Curves Stratified By Risk Categorisation; Panel B=UKRAGG Curves Stratified By Risk Categorisation; Panel C=WTCCC Curves Stratified By Risk Categorisation and Ever-Smoking Status; Panel D=UKRAGG Curves Stratified By Risk Categorisation and Ever-Smoking Status;  $\Delta$ =change in onset age;  $\Delta_m$ =maximum change in onset age across strata. doi:10.1371/journal.pgen.1003808.g005

conferred by different genetic components. We have also shown it is possible to predict the age of RA onset, using information on HLA and smoking to identify those at risk of younger and older onset RA, respectively. Whilst our novel modelling approach, which uses computer simulation-based categorisation alongside a greater number of HLA alleles, significantly improves upon the discriminative abilities of existing models [26], [27] it remains unsuitable for population screening with only a minority at significantly increased lifetime risks for RA.

Our approach provides some potential advantages over existing RA prediction modelling [26], [27]. Firstly, by using a simulated population to generate risk profiles we do not require an entire population of real-life data to stratify risks. In contrast existing approaches categorise wGRS scores using their Gaussian distribution in control groups. Secondly, our CI-based approach considers the precision with which risk factor effect sizes are known when classifying risk; this prevents classifying people high-risk if their risk is imprecisely known. Thirdly, our models provide greater discrimination: the highest AUC for existing clinical-genetic models in discerning ACPA-positive RA from controls is 0.752; the highest AUC for our clinical-genetic model is 0.857.

SNPs provided only minor improvements in prediction, highlighting the limitations of genome-wide association study (GWAS) derived data in this field. Although GWAS-established SNPs have helped identify cellular pathways relevant to RA pathogenesis [51] their modest effect sizes limit their predictive utility. It has been proposed that the missing heritability of RA may reflect the involvement of rare variants of large effect sizes or structural variants [52]. Alternative genotyping technologies such as next-generation sequencing may identify these variants, although only loci with large effect sizes will substantially improve prediction modelling.

Although individuals with clinically relevant increased lifetime risks (such as 86%) for RA were identified there was, overall, only a minority of individuals at a significantly elevated risk: 7% of ACPA-positive individuals had lifetime risks of 22% or more when evaluated using all available risk factors. Therefore despite high AUCs our modelling is unsuitable for population level screening. However, if its use was targeted to groups with *a priori* increased risks, such as first degree relatives of RA probands [53–55], then a substantially greater proportion of very high-risk individuals might be identified.

Individuals classified high-risk by our HLA model were more likely to develop RA at a younger age. This finding - mainly attributable to the \*04:01 allele - is supported by existing literature. Hellier et al reported a higher frequency of \*04 RA associated alleles in YORA (present in 52% of 262 RA cases with onset age <60) compared with elderly onset RA (present in 37%) of 60 cases with onset age >60; P=0.045) [18]. Similarly, Wu et al identified a significantly younger age of onset in Caucasian RA patients carrying shared epitope encoding \*04 alleles (P = 0.0003) [19]. Other studies report positive correlations between YORA and shared epitope alleles [25], [56]. Our finding of ever-smoking associating with older onset RA is less established. It has only been examined in three relatively small studies, with contrasting outcomes: one study reported a significant relationship between smoking at disease onset and a younger onset age [57]; one reported a younger onset age in current vs. never-smokers (although ex-smokers had older onset RA in comparison to both these groups) [58]; the final study found no association [59]. Our findings - demonstrated in 2,323 individuals across two independent datasets - are biologically plausible. As risk genotypes are present from birth they can exert their effects on disease risk throughout an individual's lifetime; therefore possessing high-risk *HLA-DRB1* alleles predisposes to RA at a younger age. In contrast the risk of RA increases as more cigarettes are smoked [60], [61] and smokers are exposed to more cigarettes as they age; therefore smokers are more likely to develop RA as they get older because they have been exposed to more cigarettes and thus smoking associates with older onset RA. This logic also explains why eversmoking associates with older onset RA in both men and women, with heavy smoking a risk factor for RA in both genders [5]. We were, however, unable to incorporate heavy smoking in our prediction modelling due to a paucity of data on smoking pack-years in WTCCC/UKRAGG.

We incorporated many genetic risk factors in our modelling but included only one environmental risk factor, smoking. This reflects uncertainty regarding relevant environmental risks alongside limited environmental data within current genetic datasets. Although many environmental factors are linked to RA their associations are usually identified in case-control studies, which are subject to multiple biases, rather than cohort studies. Examples include alcohol consumption [36], parity [62], [63] and oral contraceptive pill use [64]. Better characterisation of environmental risks will enhance predictive modelling.

Our modelling has several limitations. Firstly, WTCCC participants were included in the meta-analyses that we obtained our genetic risk loci data from; however WTCCC comprised only a proportion of the meta-analyses datasets (20% of the HLA meta-analysis; 29% of the SNP meta-analysis) and our findings were independently replicated in UKRAGG. Secondly, missing data meant the number of individuals included in each model fell as more risk factors were included; this is particularly seen in models incorporating smoking. Thirdly, due to marked heterogeneity in published data on gene-gene/gene-environment interactions we assumed independence between these factors despite known interactions existing between the shared epitope alleles and *PTPN22* and smoking [6], [7], [29], [37], [38].

Improving RA prediction requires better clarification of its genetic and environmental risk factors. Identifying risk factors with large effect sizes of known precision will most enhance prediction modelling. This could be facilitated through fine-mapping studies that better tag causal variants [65] alongside prospective cohort studies examining environmental risk factors in RA cases subdivided by ACPA status, with increasing evidence that risks differ between these serological subsets [36], [66]. It is, however, unlikely that identifying such risk factors will substantially increase the proportion of individuals with clinically relevant high disease risks. We therefore consider that prediction modelling requires evaluation in *a priori* higher risk groups. In this context it may identify sufficient numbers of very high-risk individuals, facilitating a better understanding of pre-RA immunopathology and enabling the assessment of primary prevention strategies.

# **Supporting Information**

**Figure S1** REGENT stage 1- simulation of general population risk profiles: example using a model incorporating 3 SNPs. (TIFF)

**Figure S2** Main findings of each prediction model. (TIFF)

**Table S1** Risk categorisation results for models incorporating smoking in females. Data are number (%) unless stated otherwise; <sup>a</sup> = AUCs calculated using ACPA-positive cases. (DOCX)

**Table S2** Proxy SNPs used in modelling. a = proxy SNP obtained using 1,000 Genomes CEU population panel [39]; b = proxy SNP obtained using HapMap release 22 CEU population panel [39]; c = proxy SNP obtained using Ricopili (Broad Institute, Boston, USA) from the GWAS meta-analysis of RA risk (http://www.broadinstitute.org/mpg/ricopili/). (DOCX)

**Table S3** Two-digit, four-digit and mixed-digit hla prediction model results, showing similar discriminative abilities. Data are number (%) unless stated otherwise; Sero+=seropositive RA; ACPA+=ACPA-positive RA; AUCs calculated using ACPApositive cases; two-digit model evaluated individuals with all available HLA data collapsed down to two-digit resolution. (DOCX)

#### References

- Wolfe F, Hawley DJ (1998) The longterm outcomes of rheumatoid arthritis: Work disability: a prospective 18 year study of 823 patients. J Rheumatol 25: 2108–2117.
- Michaud K, Messer J, Choi HK, Wolfe F (2003) Direct medical costs and their predictors in patients with rheumatoid arthritis: a three-year study of 7,527 patients. Arthritis Rheum 48: 2750–2762.
- Scott DL, Wolfe F, Huizinga TWJ (2010) Rheumatoid arthritis. Lancet 376: 1094–1108.
- Eyre S, Bowes J, Diogo D, Lee A, Barton A, et al. (2012) High-density genetic mapping identifies new susceptibility loci for rheumatoid arthritis. Nat Genet 44: 1336–1340.
- Sugiyama D, Nishimura K, Tamaki K, Tsuji G, Nakazawa T, et al. (2010) Impact of smoking as a risk factor for developing rheumatoid arthritis: a metaanalysis of observational studies. Ann Rheum Dis 69: 70–81.
- Pedersen M, Jacobsen S, Garred P, Madsen HO, Klarlund M, et al. (2007) Strong combined gene-environment effects in anti-cyclic citrullinated peptidepositive rheumatoid arthritis: a nationwide case-control study in Denmark. Arthritis Rheum 56: 1446–1453.
- Kallberg H, Padyukov L, Plenge RM, Ronnelid J, Gregersen PK, et al. (2007) Gene-gene and gene-environment interactions involving HLA-DRB1, PTPN22, and smoking in two subsets of rheumatoid arthritis. Am J Hum Genet 80: 867– 875.
- Schulze MB, Weikert C, Pischon T, Bergmann MM, Al-Hasani H, et al. (2009) Use of multiple metabolic and genetic markers to improve the prediction of type 2 diabetes: the EPIC-Potsdam Study. Diabetes Care 32: 2116–2119.
- Talmud PJ, Hingorani AD, Cooper JA, Marmot MG, Brunner EJ, et al. (2010) Utility of genetic and non-genetic risk factors in prediction of type 2 diabetes: Whitehall II prospective cohort study. BMJ 340: b4838.
- Paynter NP, Chasman DI, Pare G, Buring JE, Cook NR, et al. (2010) Association between a literature-based genetic risk score and cardiovascular events in women. JAMA 303: 631–637.
- Gerlag DM, Raza K, van Baarsen LGM, Brouwer E, Buckley CD, et al. (2012) EULAR recommendations for terminology and research in individuals at risk of rheumatoid arthritis: report from the Study Group for Risk Factors for Rheumatoid Arthritis. Ann Rheum Dis 71: 638–641.
- Nielen MMJ, van Schaardenburg D, Reesink HW, van de Stadt RJ, van der Horst-Bruinsma IE, et al. (2004) Specific autoantibodies precede the symptoms of rheumatoid arthritis: a study of serial measurements in blood donors. Arthritis Rheum 50: 380–386.
- Bos WH, Dijkmans BAC, Boers M, van de Stadt RJ, van Schaardenburg D (2010) Effect of dexamethasone on autoantibody levels and arthritis development in patients with arthralgia: a randomised trial. Ann Rheum Dis 69: 571–574.
- Verstappen SM, McCoy MJ, Roberts C, Dale NE, Hassell AB, et al. (2010) Beneficial effects of a 3-week course of intramuscular glucocorticoid injections in patients with very early inflammatory polyarthritis: results of the STIVEA trial. Ann Rheum Dis 69: 503–509.
- van Dongen H, van Aken J, Lard LR, Visser K, Ronday HK, et al. (2007) Efficacy of methotrexate treatment in patients with probable rheumatoid arthritis: a double-blind, randomized, placebo-controlled trial. Arthritis Rheum 56: 1424–1432.
- Emery P, Durez P, Dougados M, Legerton CW, Becker JC, et al. (2010) Impact of T-cell costimulation modulation in patients with undifferentiated inflammatory arthritis or very early rheumatoid arthritis: a clinical and imaging study of abatacept (the ADJUST trial). Ann Rheum Dis 69: 510–516.
- Lajas C, Abasolo L, Bellajdel B, Hernandez-Garcia C, Carmona L, et al. (2003) Costs and predictors of costs in rheumatoid arthritis: a prevalence-based study. Arthritis Rheum 49: 64–70.
- Hellier JP, Eliaou JF, Daures JP, Sany J, Combe B (2001) HLA-DRB1 genes and patients with late onset rheumatoid arthritis. Ann Rheum Dis 60: 531–533.
- Wu H, Khanna D, Park G, Gersuk V, Nepom GT, et al. (2004) Interaction between RANKL and HLA-DRB1 genotypes may contribute to younger age at

**Table S4** Prediction model results: risk categorisation. Data are number (%); Sero+=seropositive RA; ACPA+=ACPA-positive RA.

(DOCX)

#### Acknowledgments

We are grateful to all patients and staff contributing to WTCCC and UKRAGG.

#### **Author Contributions**

Conceived and designed the experiments: ICS SDS SS APC CML. Performed the experiments: ICS SDS RT. Analyzed the data: ICS SDS RT PF CML. Contributed reagents/materials/analysis tools: SS AH SE AWM AGW LJH PW AB JW. Wrote the paper: ICS SDS SS APC CML.

onset of seropositive rheumatoid arthritis in an inception cohort. Arthritis Rheum 50: 3093–3103.

- Jaraquemada D, Ollier W, Awad J, Young A, Silman A, et al. (1986) HLA and rheumatoid arthritis: a combined analysis of 440 British patients. Ann Rheum Dis 45: 627–636.
- MacGregor A, Ollier W, Thomson W, Jawaheer D, Silman A (1995) HLA-DRB1\*0401/0404 genotype and rheumatoid arthritis: increased association in men, young age at onset, and disease severity. J Rheumatol 22: 1032–1036.
- 22. Chen Y, Mattey DL (2012) Age at onset of rheumatoid arthritis: association with polymorphisms in the vascular endothelial growth factor A(VEGFA) gene and an intergenic locus between matrix metalloproteinase (MMP) 1 and 3 genes. Clin Exp Rheumatol 30: 894–898.
- Tan W, Wu H, Zhao J, Derber LA, Lee DM, et al. (2010) A functional RANKL polymorphism associated with younger age at onset of rheumatoid arthritis. Arthritis Rheum 62: 2864–2875.
- 24. Steer S, Lad B, Grumley JA, Kingsley GH, Fisher SA (2005) Association of R602W in a protein tyrosine phosphatase gene with a high risk of rheumatoid arthritis in a British population: evidence for an early onset/disease severity effect. Arthritis Rheum 52: 358–360.
- Karlson EW, Chibnik LB, Cui J, Plenge RM, Glass RJ, et al. (2008) Associations between human leukocyte antigen, PTPN22, CTLA4 genotypes and rheumatoid arthritis phenotypes of autoantibody status, age at diagnosis and erosions in a large cohort study. Ann Rheum Dis 67: 358–363.
- Karlson EW, Chibnik LB, Kraft P, Cui J, Keenan BT, et al. (2010) Cumulative association of 22 genetic variants with seropositive rheumatoid arthritis risk. Ann Rheum Dis 69: 1077–1085.
- Chibnik LB, Keenan BT, Cui J, Liao KP, Costenbader KH, et al. (2011) Genetic risk score predicting risk of rheumatoid arthritis phenotypes and age of symptom onset. PLoS ONE [Electronic Resource] 6: e24380.
- Wellcome Trust Case Control Consortium (2007) Genome-wide association study of 14,000 cases of seven common diseases and 3,000 shared controls. Nature 447: 661–678.
- 29. Morgan AW, Thomson W, Martin SG, Yorkshire Early Arthritis Register Consortium, Carter AM, et al. (2009) Reevaluation of the interaction between HLA-DRB1 shared epitope alleles, PTPN22, and smoking in determining susceptibility to autoantibody-positive and autoantibody-negative rheumatoid arthritis in a large UK Caucasian population. Arthritis Rheum 60: 2565–2576.
- Arnett FC, Edworthy SM, Bloch DA, McShane DJ, Fries JF, et al. (1988) The American Rheumatism Association 1987 revised criteria for the classification of rheumatoid arthritis. Arthritis Rheum 31: 315–324.
- MacGregor AJ, Snieder H, Rigby AS, Koskenvuo M, Kaprio J, et al. (2000) Characterizing the quantitative genetic contribution to rheumatoid arthritis using data from twins. Arthritis Rheum 43: 30–37.
- Goddard GHM, Lewis CM (2010) Risk categorization for complex disorders according to genotype relative risk and precision in parameter estimates. Genet Epidemiol 34: 624–632.
- Crouch DJ, Goddard GH, Lewis CM (2013) REGENT: a risk assessment and classification algorithm for genetic and environmental factors. Eur J Hum Genet 21: 109–111.
- Stahl EA, Raychaudhuri S, Remmers EF, Xie G, Eyre S, et al. (2010) Genomewide association study meta-analysis identifies seven new rheumatoid arthritis risk loci. Nat Genet 42: 508–514.
- Raychaudhuri S, Sandor C, Stahl EA, Freudenberg J, Lee H-S, et al. (2012) Five amino acids in three HLA proteins explain most of the association between MHC and seropositive rheumatoid arthritis. Nat Genet 44: 291–296.
- Scott IC, Tan R, Stahl D, Steer S, Lewis CM, et al. (2013) The Protective Effect Of Alcohol On Developing Rheumatoid Arthritis: A Systematic Review And Meta-Analysis. Rheumatology (Oxford) [Epub ahead of print].
- Kallberg H, Ding B, Padyukov L, Bengtsson C, Ronnelid J, et al. (2011) Smoking is a major preventable risk factor for rheumatoid arthritis: estimations of risks after various exposures to cigarette smoke. Ann Rheum Dis 70: 508–511.

- Karlson EW, Chang SC, Cui J, Chibnik LB, Fraser PA, et al. (2010) Geneenvironment interaction between HLA-DRB1 shared epitope and heavy cigarette smoking in predicting incident rheumatoid arthritis. Ann Rheum Dis 69: 54–60.
- Johnson AD, Handsaker RE, Pulit SL, Nizzari MM, O'Donnell CJ, et al. (2008) SNAP: a web-based tool for identification and annotation of proxy SNPs using HapMap. Bioinformatics 24: 2938–2939.
- Metz CE (1978) Basic principles of ROC analysis. Seminars in Nuclear Medicine 8: 283–298.
- Lu Q, Elston RC (2008) Using the optimal receiver operating characteristic curve to design a predictive genetic test, exemplified with type 2 diabetes. Am J Hum Genet 82: 641–651.
- DeLong ER, DeLong DM, Clarke-Pearson DL (1988) Comparing the areas under two or more correlated receiver operating characteristic curves: a nonparametric approach. Biometrics 44: 837–845.
- Robin X, Turck N, Hainard A, Tiberti N, Lisacek F, et al. (2011) pROC: an open-source package for R and S+ to analyze and compare ROC curves. BMC Bioinformatics 12: 77.
- 44. Symmons D, Turner G, Webb R, Asten P, Barrett E, et al. (2002) The prevalence of rheumatoid arthritis in the United Kingdom: new estimates for a new century. Rheumatology (Oxford) 41: 793–800.
- Zhang J, Yu KF (1998) What's the relative risk? A method of correcting the odds ratio in cohort studies of common outcomes. JAMA 280: 1690–1691.
- Crowson CS, Matteson EL, Myasoedova E, Michet CJ, Ernste FC, et al. (2011) The lifetime risk of adult-onset rheumatoid arthritis and other inflammatory autoimmune rheumatic diseases. Arthritis Rheum 63: 633–639.
- Hess KR (1995) Graphical methods for assessing violations of the proportional hazards assumption in Cox regression. Stat Med 14: 1707–1723.
- Huizinga TWJ, Amos CI, van der Helm-van Mil AHM, Chen W, van Gaalen FA, et al. (2005) Refining the complex rheumatoid arthritis phenotype based on specificity of the HLA-DRB1 shared epitope for antibodies to citrullinated proteins. Arthritis Rheum 52: 3433–3438.
- Orozco G, Pascual-Salcedo D, Lopez-Nevot MA, Cobo T, Cabezon A, et al. (2008) Auto-antibodies, HLA and PTPN22: susceptibility markers for rheumatoid arthritis. Rheumatology (Oxford) 47: 138–141.
- Klareskog L, Stolt P, Lundberg K, Kallberg H, Bengtsson C, et al. (2006) A new model for an etiology of rheumatoid arthritis: smoking may trigger HLA-DR (shared epitope)-restricted immune reactions to autoantigens modified by citrullination. Arthritis Rheum 54: 38–46.
- Plenge RM (2009) Recent progress in rheumatoid arthritis genetics: one step towards improved patient care. Curr Opin Rheumatol 21: 262–271.
- de Vries RRP, van der Woude D, Houwing JJ, Toes REM (2011) Genetics of ACPA-positive rheumatoid arthritis: the beginning of the end? Ann Rheum Dis 70 Suppl 1: i51–54.

- Wolfe F, Kleinheksel SM, Khan MA (1988) Prevalence of familial occurrence in patients with rheumatoid arthritis. Br J Rheumatol 27 Suppl 2: 150–152.
- Deighton CM, Wentzel J, Cavanagh G, Roberts DF, Walker DJ (1992) Contribution of inherited factors to rheumatoid arthritis. Ann Rheum Dis 51: 182–185.
- Koumantaki Y, Giziaki E, Linos A, Kontomerkos A, Kaklamanis P, et al. (1997) Family history as a risk factor for rheumatoid arthritis: a case-control study. J Rheumatol 24: 1522–1526.
- Fries JF, Wolfe F, Apple R, Erlich H, Bugawan T, et al. (2002) HLA-DRB1 genotype associations in 793 white patients from a rheumatoid arthritis inception cohort: frequency, severity, and treatment bias. Arthritis Rheum 46: 2320–2329.
- Hutchinson D, Shepstone L, Moots R, Lear JT, Lynch MP (2001) Heavy cigarette smoking is strongly associated with rheumatoid arthritis (RA), particularly in patients without a family history of RA. Ann Rheum Dis 60: 223–227.
- Papadopoulos NG, Alamanos Y, Voulgari PV, Epagelis EK, Tsifetaki N, et al. (2005) Does cigarette smoking influence disease expression, activity and severity in early rheumatoid arthritis patients? Clin Exp Rheumatol 23: 861–866.
- Diaz FJ, Rojas-Villarraga A, Salazar JC, Iglesias-Gamarra A, Mantilla RD, et al. (2011) Anti-CCP antibodies are associated with early age at onset in patients with rheumatoid arthritis. Joint, Bone, Spine: Revue du Rhumatisme 78: 175– 178.
- Costenbader KH, Feskanich D, Mandl LA, Karlson EW (2006) Smoking intensity, duration, and cessation, and the risk of rheumatoid arthritis in women. Am J Med 119: 503.c501–509.
- 61. Stolt P, Bengtsson C, Nordmark B, Lindblad S, Lundberg I, et al. (2003) Quantification of the influence of cigarette smoking on rheumatoid arthritis: results from a population based case-control study, using incident cases. Ann Rheum Dis 62: 835–841.
- Karlson EW, Mandl LA, Hankinson SE, Grodstein F (2004) Do breast-feeding and other reproductive factors influence future risk of rheumatoid arthritis? Results from the Nurses' Health Study. Arthritis Rheum 50: 3458–3467.
- Guthrie KA, Dugowson CE, Voigt LF, Koepsell TD, Nelson JL (2010) Does pregnancy provide vaccine-like protection against rheumatoid arthritis? Arthritis Rheum 62: 1842–1848.
- Spector TD, Hochberg MC (1990) The protective effect of the oral contraceptive pill on rheumatoid arthritis: an overview of the analytic epidemiological studies using meta-analysis. J Clin Epidemiol 43: 1221–1230.
- 65. Scott IC, Steer S, Lewis CM, Cope AP (2011) Precipitating and perpetuating factors of rheumatoid arthritis immunopathology: linking the triad of genetic predisposition, environmental risk factors and autoimmunity to disease pathogenesis. Best Pract Res Clin Rheumatol 25: 447–468.
- 66. Pedersen M, Jacobsen S, Klarlund M, Pedersen BV, Wiik A, et al. (2006) Environmental risk factors differ between rheumatoid arthritis with and without auto-antibodies against cyclic citrullinated peptides. Arthritis Res Ther 8: R133.